CSE 521: The Design and Analysis of Algorithms

Assignmen t #5

Ma y 21, 1997

due: W ednesda y , Ma y 28

1. Exercise 15.1-5 on page 286.

2. Exercise 15.2-3 on page 289.

3. (a) Exercis e 15.2-2 on page 289.

(b)-(e) Problem 14-2, parts (b)-(e), on page 278.

I ha v e purp osely substitute d Exercise 15.2-2 for part (a) of Problem

14-2, since w e did not co v er the details of RB-Delete.

4. Exercise 15.3-7 on page 295. The rectangles include their b oundaries,

and \o v erlap" means \ha v e a nonempt y in tersec tion", as usual. Justify

the running time of y our algorithm.

5. Exercise 20.2-10 on page 417.

6. Problem 20-2 on page 418. Include a careful explanation of ho w y ou

implemen t the test in line 9, without the necessi t y of a separate Union-

Find data structure. (Hin t: augmen t the binomial heap no des with an

additional �eld.)